Author’s response to reviews

Title: Healthy Eating Index versus Alternate Healthy Eating Index in relation to Diabetes status and health markers in US adults

Authors:
Afnan Al Ibrahim (afnan_alibrahim@hotmail.com)
Robert Jackson (bojack@umd.edu)

Version: 1 Date: 25 Oct 2018

Author’s response to reviews:

Dear Editors of Nutrition Journal,

We wish to submit a manuscript for consideration of publication in the Nutrition Journal. The manuscript is entitled "Healthy Eating Index versus Alternate Healthy Eating Index in relation to Diabetes status and health markers in US adults". We verify that this work is an original research and that it has not been published elsewhere, nor it is currently under consideration elsewhere for publication.

As requested, the STROBE-Nut checklist are referenced as Superscript at the beginning of subtitles and sentences within the manuscript, and a flow diagram is included in the main body of the manuscript.

We verify that the details on the revised version are correct, and we agreed to authorship and order of authorship for this manuscript. We verify that all authors have the appropriate permissions and rights to the reported data.

Please address all correspondence concerning this manuscript to Dr. Afnan Al-Ibrahim (email: afnan_alibrahim@hotmail.com). Thank you for your consideration of this manuscript.

Sincerely,

Manuscript Authors